Author's response to reviews

Title: Genotyping and Drug resistance patterns of M. tuberculosis strains in Pakistan

Authors:

Mahnaz Tanveer (mahnaz.tanveer@aku.edu)
Zahra Hasan (maqboola.dojki@aku.edu)
Amna R Siddiqui (rehana.siddiqui@aku.edu)
Ahsa Ali (asho.ali@aku.edu)
Akbar Kanji (akbar.kanji@aku.edu)
Solomon Ghebremicheal (solomon.ghebremicheal@smi.ki.se)
Rumina Hasan (rumina.hasan@gmail.com)

Version: 6 Date: 17 December 2008

Author's response to reviews: see over
Dear Dr. Zauner,

We thank the reviewer for their valuable comments given on my manuscript entitled “Genotyping and Drug resistance patterns of M. tuberculosis strains in Pakistan.” The manuscript has been accordingly revised and the replies to the queries raised by the reviewer are attached.

Thanking you,

Yours Sincerely,
Rumina Hasan
Professor and Chairperson,
Department of Pathology and Microbiology.
The Aga Khan University
Karachi, Pakistan.

Encls: 1. Answers to reviewers comments
    2. Revised manuscript